College of Computer Science Fall 2002 Northeastern University Handout 5 COM 1721: Freshman Honors Seminar 8 October 2002

Homework 3

1. Read the article “An Introduction to Computer Chess”, available at

http://se.uwaterloo.ca/~dalopez-o/divulge/chimp.html

Bring a (typed) report on your reading to the next class, Tuesday, October 15. Your reports will be considered in your evaluation.

Reports must be very brief, with one to three sentences on each of the following items:

1. A brief description of the article.

2. Something in the reading that you found interesting; e.g., a fact or an opinion that the author expressed.

3. A response to the reading: a comment or question about the article or your opinion on the subject.

A few students would be asked to read out their reports in the class, and we will have a short discussion on the topic.